Author’s response to reviews

Title: Technical Requirements Framework of Hospital Information Systems: Design and Evaluation

Authors:
Mehrdad Farzandipour (farzandipour_m@kaums.ac.ir)
Zahra Meidani (zmeidani@yahoo.com)
Ehsan Nabovati (enabovati@gmail.com)
Monireh Sadeqi Jabali (msadeqi2005@gmail.com)
Razieh Dehghan Banadaki (razyedehghan@yahoo.com)

Version: 2 Date: 12 Feb 2020

Author’s response to reviews:

Dear Editor of BMC Medical Informatics and Decision Making Journal

Regarding paper number MIDM-D-19-00405R1

We would like to thank the dear editor and reviewers of the journal and mention that authors thought that, by this comment, the reviewer had written a statement, not a comment to be edited. That’s why no response was given to this comment. We express our apologies. The comment has been responded to as follows.

Reviewer Batista-Navarro's concern no. 3: "I think the manuscript can benefit from the authors commenting on the applicability of the proposed framework to other country settings".
The response was added with blue color to conclusion section, line 1 and 3-7, page 17:
- This framework refers to technical requirements of all HISs in general. Other items can be redefined so as to suit the requirements of other HISs. Therefore, this framework may apply to other country settings as well. Each country, depending on its degree of development and progress in the implementation of HISs, can benefit from the present framework.

Also Regarding paper number MIDM-D-19-00405R2, the requested corrections were made on the title page and authors' contribution of the manuscript and highlighted with blue color in the manuscript.

Thanking you,
Yours’ sincerely
Mehrdad Farzandipour
Corresponding Author